\documentclass[12pt]{article}

\setlength{\parindent}{0pt}
\everymath={\displaystyle}
\pagestyle{empty}
\newcommand{\answerspace}{\vfill}

\begin{document}

\textbf{Math 126: Calculus II}
\hfill\makebox[2.5in][r]{Name:\hrulefill}

\textbf{Quiz 9}\quad \textsl{November 14, 2000}
\hfill\makebox[2.5in][r]{Section:\hrulefill}

\bigskip



 

1. Find the interval of convergence of the power series:
  \[ \sum_{n=1}^\infty \frac{(x-4)^n}{n 3^n}\]
 

\answerspace

2. Find the Taylor polynomial of order 3 for $f(x) = \sin(2x)$
   at $x = 0$.


\answerspace

\end{document}

